Author's response to reviews

Title: The Top Tertile of Hematocrit Change During Hospitalization is Associated with Lower Risk of Mortality in Acute Heart Failure Patients

Authors:

Haobin Zhou (zhouhaobin021@163.com)
Tianyu Xu (ertrtyu0915@163.com)
Yuli Huang (hyuli821@163.com)
Qiong Zhan (zhanqiong2010@163.com)
Xingfu Huang (happyhxf@163.com)
Qingchun Zeng (z-qch@163.com)
Dingli Xu (dinglixu@fimmu.com)

Version: 3 Date: 19 Aug 2017

Author’s response to reviews:

1. Please clarify whether the patients were aware that their data was being used in this study, and did they consent to their information being used. If the data was anonymised, and thus consent would not be needed, please state this in the Ethical approval and consent to participate section.

Response: Thank you for your suggestion, we added this to the Ethical approval and consent to participate section: “Because the data in this study was anonymised, no patient consent was required.”.

2. Please include all authors’ email addresses on the title page.

Response: This information was added to the title page.

3. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: We have revised the manuscript according to your suggestion.
4. Please remove all supplementary files which you do not want published, as they are no longer needed at this stage.

Response: We have removed the files according to your suggestion.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: The manuscript will be uploaded as required.